Author's response to reviews

Title: Deoxycholate induces COX-2 expression via Erk1/2-, p38-MAPK and AP-1-dependent mechanisms in esophageal cancer cells

Authors:

Eileen Looby (loobye@tcd.ie)
Mohamed M. M. Abdel-Latif (abdellmm@tcd.ie)
Veronica Athié-Morales (amoraley@tcd.ie)
Shane Duggan (shduggan@tcd.ie)
Aideen Long (longai@tcd.ie)
Dermot Kelleher (dermot.kelleher@tcd.ie)

Version: 3 Date: 25 May 2009

Author's response to reviews: see over
25-05-2009

Editor-in-Chief
Melissa Norton, MD
BMC Cancer Journal

MS: 7660464622306499

Prof. Melissa Norton
Thank you for your acceptance letter of our manuscript (MS: 7660464622306499) titled: Deoxycholate induces COX-2 expression via Erk1/2-, p38-MAPK and AP-1-dependent mechanisms in esophageal cancer cells.

The following sections (Competing Interests and Authors' contributions) have been completed at the end of manuscript text, preceding the 'Acknowledgements'.

The revised manuscript also conforms to the journal style.

Yours sincerely,
Dr. Aideen Long
Department of Clinical Medicine and Institute Molecular Medicine, Trinity Centre for Health Sciences, Trinity College Dublin, St. James’s Hospital, Dublin 8, Ireland
Fax: +353-1-4542043
E-mail: longai@tcd.ie